Author’s response to reviews

Title: Waiting for attention and care: birthing accounts of women in rural Tanzania who developed obstetric fistula as an outcome of labour

Authors:

Lilian T Mselle (nakutz@yahoo.com)
Thecla W Kohi (thecla.kohi@gmail.com)
Abu Mvungi (akambaga@yahoo.com)
Bjørg Evjen-Olsen (Bjorg.olsen@cih.uib.no)
Karen-Marie Moland (Karen.Marie.Moland@hib.no)

Version: 8 Date: 16 October 2011

Author’s response to reviews: see over
The Editor  
BMC Pregnancy and Childbirth  

Re: Resubmission of Manuscript no. 1968925534530759  

On behalf of my colleagues, I thank you and the editorial team for accepting our manuscript for publication in the BMC Pregnancy and Childbirth journal.  

We are pleased to resubmit, after correcting the few typographic errors on the manuscript titled “Waiting for attention and care: Birthing accounts of women in rural Tanzania who developed obstetric fistula as an outcome of labour” for publication in your journal.  

Respectfully,  

Lilian Teddy Mselle  
Email: nakutz@yahoo.com